Author's response to reviews

Title: Status of respectful and non-abusive care during facility-based childbirth in a hospital and health centers in Addis Ababa, Ethiopia

Authors:

Anteneh Asefa (antex98@yahoo.com)
Delayehu Bekele (delayehu@gmail.com)

Version: 3
Date: 13 March 2015

Author's response to reviews: see over
12 March 2015

To: Editor-in-Chief
   Reproductive Health

Dear editor, on behalf of the authors I, the corresponding author of this manuscript would like to pass our heartfelt thanks to you for the feedbacks and constructive suggestions which made the manuscript at its current version. In this third version of our manuscript, we have incorporated your comments and suggestions accordingly. Here under is the list of major changes made.

- The manuscript is edited by scientific English editors (Nextgenediting) and another scientific writer, Aku Kwamie. I hope you will be satisfied this time as significant improvement is made.
- The discussion part is synchronized as per your suggestion.
- We have referred to the author’s instruction of the journal and corrected the “references” section accordingly.

For detail information on the changes made, please refer to the manuscript which is track changed.

Sincerely,

Anteneh Asefa
Corresponding author